Author’s response to reviews

Title: A moderate 500-m treadmill walk for estimating peak oxygen uptake in men with NYHA class I-II heart failure and mid-range ventricular dysfunction

Authors:
Gianni Mazzoni (gianni.mazzoni@unife.it)
Biagio Sassone (b.sassone@ausl.fe.it)
Giovanni Pasanisi (g.pasanisi@ausl.fe.it)
Jonathan Myers (DRJ993@aol.com)
Simona Mandini (mndsmn@unife.it)
Stefano Volpato (stefano.volpato@unife.it)
Francesco Conconi (cnf@unife.it)
Giorgio Chiaranda (Giorgio.Chiaranda@regione.emilia-romagna.it)
Giovanni Grazzi (giovanni.grazzi@unife.it)

Version: 5 Date: 28 Feb 2018

Author’s response to reviews:

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Therefore, we would appreciate if you try to minimise direct repetition of large sections of previously published text. For more information regarding where this overlap occurs please see the attached screenshot.

Reply: Dear Dr. Dickens, following your suggestion we have modified the method section. The new version is attached.